Author's response to reviews

Title: Weight change following breast cancer: evidence from a prospective, population-based, breast cancer cohort study

Authors:

Dimitrios Vagenas (dimitrios.vagenas@qut.edu.au)
Tracey DiSipio (t.disipio@qut.edu.au)
Diana Battistutta (d.battistutta@qut.edu.au)
Wendy Demark-Wahnefried (demark@uab.edu)
Sheree Rye (sarye@qut.edu.au)
John Bashford (jbashford@hoca.com.au)
Chris Pyke (c_pyke@mc.mater.org.au)
Christobel Saunders (christobel.saunders@uwa.edu.au)
Sandra C Hayes (sc.hayes@qut.edu.au)

Version: 2 Date: 6 May 2014

Author's response to reviews:

As requested, this manuscript has been revised to include line numbers.